# EVOLVING ASPIRATIONS AND COOPERATION

Rajeeva Karandikar, Dilip Mokherjee, Debraj Ray and Fernando Vega-Redondo<sup>\*</sup>

WP-AD 96-06

<sup>&</sup>lt;sup>\*</sup> Rajeeva Karandikar: Indian Statistical Institute, Dilip Mokherjee: Boston University and Indian Statistical Institute. Debraj Ray: Boston University and Instituto de Análisis Económico (CSIC). Fernando Vega-Redondo: Universidad de Alicante and Instituto Valenciano de Investigaciones Económicas.

### EVOLVING ASPIRATIONS AND COOPERATION

Rajeeva Karandikar, Dilip Mokherhee, Debraj Ray & Fernando Vega-Redondo

### ABSTRACT

A model of "satisficing" behavior in the repeated Prisoners Dilemma is studied. Each player has an *aspiration* at each date, and takes an action. [S]he switches from the action played in the previous period only if the achieved payoff fell below the aspiration level (with a probability that depends on the shortfall). Aspirations are updated in each period, according to payoff experience in the previous period. In addition, aspirations are subjected to random perturbations around the going level, with a small "tremble" probability. For sufficiently slow updating of aspirations, and small tremble probability, it is shown that in the long run both players cooperate most of the time.

**KEYWORDS:** Cooperation; Aspirations; Learning.

### 1 Introduction

Consider a Prisoner's Dilemma played repeatedly by two 'satisficing' players. Each player has an *aspiration* at each date, and takes an action. [S]he switches from the action played in the previous period only if the achieved payoff fell below the aspiration level (with a probability that depends on the shortfall). Aspirations are updated in each period, depending on the divergence of achieved payoffs from aspirations in the previous period.

The aspiration-based process is characterized by two specific features: (i) *inertia*: every action is repeated with at least a certain probability bounded away from zero; and (ii) *experimentation*: with a small probability, players' aspiration levels experience small random trembles around the "going" aspiration, thus preventing them from being perpetually "satisfied" with any given action. We examine the long run outcomes that are induced by vanishingly small tremble probabilities. This paper therefore builds on the work of Bendor, Mookherjee and Ray [1992], in which a model of "consistent" aspirationsbased learning was introduced.<sup>5</sup>

We make precise and prove the following result (Theorem 1 below):

If the speed of updating aspiration levels is sufficiently slow, then the outcome in the long run involves both players cooperating "most of the time".

The model therefore describes an adaptive learning process where individuals not only cooperate, but play strictly dominated strategies of the stage game for "most of the time". While players may (and occasionally do) profit by deviating from cooperative behavior, the dynamics of the process ultimately lead back to mutual cooperation.

Section 2 introduces the model, and some of its preliminary properties Section 3 presents the main results. Section 4 contains an informal discussion of some of the results. Section 5 discusses related literature All proofs are in Section 6

# 2 The Model

Consider a Prisoner's Dilemma:

	С	D
С	$(\sigma, \sigma)$	$(0,\theta)$
D	$(\theta, 0)$	$(\delta, \delta)$

where  $\theta > \sigma > \delta > 0$ 

<sup>&</sup>lt;sup>5</sup>That paper did not consider the updating of aspirations within a game. For discussion of this and related literature, see Section 5.

Player 1's state at date t is given by an action  $A_t \in \{C, D\}$ , and an aspiration level  $\alpha_t$ , which is a real number. The corresponding objects for player 2 are given by  $B_t$  and  $\beta_t$ . A state is the pair made up of player 1's state and player 2's state.

The state at date t determines payoffs  $\pi_t^1$  and  $\pi_t^2$  for the two players We now describe the *updating* of each player's state.

We describe the rule followed by player 1; an analogous rule is followed by player 2. There are two features. First, actions are updated as follows: if  $\pi_t^1 \ge \alpha_t$ , then player 1 is satisfied, and  $A_t = A_{t+1}$ . Otherwise player 1 is disappointed, so [s]he switches action  $(A_{t+1} \ne A_t)$  with probability 1-p, where p is an indicator of inertia. It is assumed that p is a nonincreasing function of the extent of disappointment  $(\alpha_t - \pi_t^1)$ , satisfying:

- 1. p = 1 if  $\alpha_t \pi_t^1 \leq 0$ , and
- 2.  $p \in (\tilde{p}, 1)$  otherwise, for some  $\tilde{p} \in (0, 1)$ .
- 3. p is continuous and the rate at which it falls is bounded, i.e., for all  $x > 0, 1-p(x) \le Mx$  for some  $M < \infty$ .<sup>6</sup>

Figure 1 describes *p*. In words, for any given positive degree of disappointment, the player will switch his action with positive probability. However, the probability of *not* switching is bounded away from zero.

Second, with respect to the *updating of aspirations*, it is convenient to first consider the case without any "trembles".

#### 2.1 The Model without Trembles

Aspirations are updated as an average of the aspiration level and the achieved payoff at the previous play. For player 1, this yields

$$\alpha_{t+1} = \lambda \alpha_t + (1 - \lambda) \pi_t^1, \tag{1}$$

where  $\lambda \in (0, 1)$  may be thought of as a *persistence* parameter, assumed equal for both players for simplicity. A parallel equation applies to player 2.

The updating rules (without trembles) define a Markov process over the state space  $\{C, D\}^2 \times \mathbb{R}^2$ . This process will be denoted P, and will be referred to as the *untrembled* process

Given any action pair (A, B), let the corresponding *pure strategy state* (pss) refer to the state where this action pair is played with aspiration levels exactly equal to the

<sup>&</sup>lt;sup>6</sup>This last set of conditions is inessential for our main result, though employed in the proof As will be discussed in Section 4, modified arguments apply in the case where the inertial probability is discontinuous at the point of zero disappointment, i.e., is bounded away from 1 as well as 0 for any positive disappointment.



Figure 1: THE FUNCTION p.

achieved payoffs:  $\alpha = \pi^1(A, B), \beta = \pi^2(A, B)$ . It is clear that every pss is an absorbing state of the untrembled process: if players are satisfied with the payoffs they receive in an ongoing action pair, they have no reason to alter their actions or aspirations. Indeed, it is for this very reason that it is necessary to explore the robustness of any absorbing state by admitting the possibility of perturbations.

Before proceeding to the case of trembles, however, it is useful to settle a preliminary question: does the untrembled process always converge to some pss?

**PROPOSITION 1** From any given initial state, the untrembled process P converges almost surely to some pure strategy state.

This result is discussed further in Section 4.

### 2.2 The Model with Trembles

While the untrembled process always converges to some pss, one suspects that some of these may not be robust to small perturbations in a player's state. To model such phenomena, we introduce trembles in the formation of aspirations.<sup>7</sup>

<sup>&</sup>lt;sup>7</sup>Starting with any pss, a small upward push to a player's aspiration level will cause that player to be disappointed, and hence induced to experiment with other actions. An alternative modeling approach

With probability  $1 - \eta$ , aspirations are formed according to the deterministic rule (1), while with the remaining probability  $\eta$ , the updated deterministic aspiration  $\alpha$  is perturbed according to some density  $g(\cdot, \alpha)$ . Assume, again for simplicity, that  $\eta$  is the same for both players.

Informally, we would like small perturbations on either side of  $\alpha$  to be possible, but at the same time, uninteresting technical complexities would be introduced by allowing aspirations to wander too far from the payoffs of the game, and we want to avoid these. In what follows, we impose appropriate restrictions.

Assume, then, that there exists some compact interval  $\Lambda$  which contains all feasible payoffs such that for each  $\alpha \in \Lambda$ , the support of  $g(|\alpha)$  is contained in  $\Lambda$ , and moreover, that  $g(\alpha'|\alpha) > 0$  for all  $\alpha'$  in some nondegenerate interval around  $\alpha$  (relative to  $\Lambda$ ). Furthermore, suppose that g is continuous as a function of  $\alpha$ .

Finally, assume that all initial aspiration vectors lie in the compact region  $\Lambda^2$ , and that all perturbations are independent over time and across players.<sup>8</sup>

Denote the resulting stochastic process by  $P^{\eta}$ . A standard theorem (see, e.g., Meyn and Tweedie (1993), Theorem 16.2.5) guarantees that the process has a well defined long run outcome:

PROPOSITION 2 For  $\eta > 0$ , the process  $P^{\eta}$  converges (strongly) to a unique limit distribution  $\Pi^{\eta}$ , irrespective of the initial state.

# 3 Main Results

By Proposition 2, the introduction of trembles serves to single out a unique (though probabilistic) long-run outcome. Obviously, one is interested in the nature of the long run distribution  $\Pi^{\eta}$  when the tremble probability  $\eta$  is close to zero, since this is likely to yield a selection from the multiple long-run limits of the untrembled process.

Some preliminary steps are needed before we can state such a result precisely. To begin with, one needs to ensure that the sequence of long run distributions  $\Pi^{\eta}$  has a well-defined limit as  $\eta$  goes to zero.

**PROPOSITION 3** The sequence of distributions  $\Pi^{\eta}$  converges weakly to a distribution  $\Pi^*$ on E as  $\eta \downarrow 0$ .

Proposition 3 is a corollary of a general theorem on the long run behavior of Markov processes subjected to small stochastic perturbations, which also provides a precise characterization of the limiting distribution. Because this result may be of wider interest than

would directly allow experimentation with different actions. We suspect that the results would be the same in such an approach

<sup>&</sup>lt;sup>8</sup>As usual, a similar definition holds for player 2. It is immaterial to the argument whether the function g is the same for both players.

the specific application studied here, we provide a self-contained statement of this general theorem in Section 5 (Theorem 2). Theorem 2 also yields the following description of the limit distribution  $\Pi^*$ .

Use  $Q_i$  to denote the one step transition probability of the stochastic process, conditional on the situation where *only* player *i*'s aspiration is subjected to a tremble. Let Qdenote  $\frac{1}{2}(Q_1 + Q_2)$ . We may interpret this as the transition rule when exactly one player trembles, with both players being equally likely to tremble.

Let R denote  $P^{\infty}$ , the infinite step transition rule in the untrembled process. This is well-defined by Proposition 1. Finally, let QR denote the composition of Q and RIn words, the process QR refers to the effect of subjecting exactly one player (chosen randomly) to a tremble in her aspirations, followed by the untrembled process thereafter for ever.

# **PROPOSITION 4** The limiting distribution $II^*$ is the unique invariant distribution of the process QR

By Proposition 1, the untrembled process converges to a pss It follows that an invariant distribution of QR must be concentrated on the pss's. Proposition 4 says that QR has a unique invariant distribution, which is precisely the limit of the invariant distributions corresponding to vanishing tremble probabilities.

In words, the selected long run outcome can be obtained as the unique long run outcome of an "artificial" Markov process defined only over the four pss's, with the transition probability between pss's obtained as follows: Starting with the former pss, subject one player chosen randomly to a single tremble in aspiration to obtain a new state, from which the untrembled process is left to operate thereafter to arrive eventually at some pss.

Why do we need a precise characterization of the limit of the long run distributions, unlike previous authors? The reason is that the long run distribution will *not* generally be concentrated on a single pss. This is in sharp contrast with random matching contexts considered by Kandori, Mailath and Rob (1993) and Young (1993), where the corresponding process singles out a unique limit state.

In the informal discussion, which we relegate to Section 4, we attempt to provide a clearer intuitive explanation of this observation.

We are now in a position to state our main result

THEOREM 1 The weight  $\Pi^*(C, C)$  placed by the limiting distribution  $\Pi^*$  on the mutual cooperation pure strategy state is close to 1, for persistence parameter  $\lambda$  sufficiently close to 1. Formally,

$$\lim_{\lambda \to 1} \Pi^*(C, C) = 1.$$

# 4 Informal Discussion

The assertions thus far may be summarized as follows. First, the untrembled joint process of aspirations and actions always converges to a pure strategy steady state. Second, the trembled process is ergodic and converges to a unique invariant distribution. Third, these invariant distributions (viewed as functions of the tremble) themselves settle down to a "limit" invariant distribution. Finally — and this is the main result — the limit invariant distribution places almost all weight on the cooperative outcome, provided that the persistence parameter is sufficiently close to unity. We discuss these informally in turn.

Begin with the convergence of the untrembled process Our formal proof takes the easiest route towards establishing convergence, by exploiting a degree of inertia in the model that is built in by assumption. Specifically, given any state, there can be an infinite run on the current action pair, which would cause aspirations to converge to the corresponding payoffs.

We illustrate this by starting with the (C, D) action vector, with 2's aspiration lying above the cooperative payoff, while player 1's aspiration does not exceed this payoff What is the probability of an infinite run on (C, D) thereafter in the untrembled process, which would result in convergence to the (C, D) pss? Along such a path, player 2 would have no cause to switch away from D, so what is needed is for player 1 to stick with Cperpetually despite being disappointed at every stage. At stage t, player 1's aspiration and hence disappointment level would be  $\lambda^t \sigma$ . Hence the probability of converging to (C, D) is  $\prod_{t=1}^{\infty} p(\lambda^t \sigma)$ , which is positive if and only if

$$\sum_{t=1}^{\infty} [1 - p(\lambda^t \sigma)] < \infty.$$

a condition which is satisfied, since the left hand side equals

$$\sum_{t=1}^{\infty} [p(0) - p(\lambda^t \sigma)] \le \frac{M\lambda\sigma}{1-\lambda}.$$

A similar argument can be given for the possibility of an infinite run on any action pair, and any initial set of aspirations, as detailed in the proof of Proposition 1. Note that no assumption has been made concerning the speed at which aspirations are updated. This is an implication of our assumptions concerning inertia

Howver, it must be emphasized that while this is the easiest way to prove Proposition 1, the result also holds under weaker assumptions concerning the inertia function p, providing we restrict aspirations to not be updated too rapidly. For instance, if the inertia probability p is discontinuous at 0, so players switch with a probability bounded away from zero whenever disappointed, the infinite runs used above to secure convergence



Figure 2: ILLUSTRATION OF THEOREM 1.

are no longer possible. Here is a sketch of an alternative argument for convergence for the case where  $\lambda$  is close to 1. Suppose, contrary to the assertion of the proposition, that the untrembled process does not converge. Then it can be shown that the process must wander infinitely often through the interior of the rectangle *I* depicted in Figure 2. Now concentrate on this rectangle. Observe that while aspirations lie within this rectangle, there is a positive probability that (C, C) will be played soon thereafter. To see this, suppose first that (D, D) is played. Then both players will be disappointed so with positive probability both will switch to (C, C) If (D, C) is played, player 1 will be happy while player 2 will be disappointed, precipitating (D, D) next period with positive probability, whereupon the previous argument applies. A completely parallel argument holds for the action pair (C, D).

All that remains is to bound this probability away from zero, which can be done by applying the argument to a suitably chosen compact subrectangle of I. Now couple this observation with the infinite recurrence of I to establish convergence to (C, C). This contradicts the assumption that the untrembled process does not converge, and we are done



Figure 3: Possible convergence to (D, D) from perturbed (C, C).

The observation that the trembled process is ergodic is a standard one, familiar by now in the literature on stochastic evolution. What is of interest is that *unlike* the results in that literature, it is perfectly possible for the limit invariant distribution (as trembles vanish) to place weight on more than one pure strategy state. In part this comes from the kind of inertia-based arguments provided above. But it is important to observe, yet again that inertia is only sufficient and not necessary to generate these features. For instance, a "long aspiration cycle" can be triggered by a perturbation from the (C, C) pss, leading to ultimate convergence (with positive probability) to the (D, D) pss from below (Figure 3 depicts such a path for a particular parametric specification, albeit somewhat cryptically, by the curved arrow). These possibilities necessitate a more detailed analysis. To get a handle on which pure strategy state is likely to receive the lion's share of probability weight, we must deduce a formula for the limit invariant distribution. This is done in Theorem 2, which is stated and proved below.

This theorem is the building block on which our proof of the main result (Theorem 1 is based. We show that while the limiting distribution assigns some weight to all pss's, the weight placed on the (C, D), (D, C) or (D, D) pss's must become arbitrarily small

when aspirations adjust sufficiently slowly: the process spends "most of its time" in the vicinity of the (C, C) outcome. In this sense, mutual cooperation is the unique long run outcome.

The proof of the theorem is long and involved, requiring an assessment of the limiting probabilities of transiting from the (C, C) pss to the other pss's (in the process QR), relative to transitions in the reverse direction, as  $\lambda$  converges to 1. It is shown that the probability of transiting from any of the other pss's to the (C, C) pss is bounded away from zero for all values of  $\lambda$  close to 1, whereas the probabilities of the reverse transitions converge to zero as  $\lambda \to 1$ .

The main reason for this is that transitions following a single perturbation of the (C, C) pss to the other pss's must *necessarily* require one or both players to not switch their actions for long stretches of time, even if they are disappointed. As aspirations adjust more and more slowly, the bouts of inertia required become indefinitely large, and therefore increasingly improbable.<sup>9</sup>

On the other hand, the transition to the (C, C) pss, can be shown to be inertia-free relative to the reverse transitions described above. Showing that this is so involves some long and delicate calculations, embedded largely in Lemma 3 For instance, starting with the (D, D) pss, when player 2 experiences a positive tremble on his/her aspirations, aspirations move to a point like N in Figure 2. From that point onwards, we show that aspirations tend to drift back, with large probability, into the interior of the rectangle I, and that this argument can be made without any reliance on inertia, so that it survives even when the persistence parameter is close to unity. Intuitively, the argument is simple Imagine a point just to the left of N in Figure 2. Then player 1 has aspirations low enough so that he is satisfied with his choice of the D action irrespective of what player 2 does. Player 2, dissatisfied, will switch back between C and D. This induces a drift of aspirations to the southeast, pointing into the interior of I. What is needed is a similar result for the point N itself and points even slightly to the right of it, which necessitates a complicated analysis.

Once the process drifts back into I, we are in a familiar realm. We have already argued (see above) that from this point on, convergence to (C, C) occurs with positive probability, in a way that does not rely on inertia.

Note in conclusion that even for small tremble probabilities and speed of aspiration updating, the process does *not* converge to mutual cooperation in the long run. Cooperation simply becomes statistically dominant. Players perpetually oscillate between different action pairs, as their aspirations are occasionally subjected to trembles in different directions, with effects that last beyond the trembles. In particular, these trembles

<sup>&</sup>lt;sup>9</sup>If the inertia function p were to be discontinuous at 0, then the probability of converging to any other pss from the cooperative pss following one tremble is obviously zero, so the result is straightforward in that case.

cause them to experiment with actions different from those used in the recent past. For instance, starting with mutual cooperation, such trembles induce experimentation with defection, which leads to transitory gains from exploiting the cooperative partner. However, the partner is dissatisfied in such situations, and will switch to defection as well, which serves to "punish" the initial defector. A period of mutual defection then ensues, which tends to disappoint *both* players, inducing an eventual return to the cooperative phase. The players play a dominated strategy often not owing to a feature of the learning rule which causes them to stick to an inferior action in a single person environment, but rather owing to the nature of interaction between the decision rules chosen by the two players.

## 5 Related Literature

The model of learning considered in this paper presumes a limited form of rationality, where players need not know the structure of the game, or the opponents' previous actions; nor do they have to be able to solve maximization problems. In the terminology of Selten (1991), such models represent "stimulus" or "reinforcement" rather than "belief" learning. They originated in the mathematical psychology literature (see Bush and Mosteller (1955)), and have received a certain degree of support in laboratory experiment situations involving human subjects (see Suppes and Atkinson (1960), Selten and Stoecker (1986), Roth and Erev (1993), Mookherjee and Sopher (1994, 1995)). Other recent explorations of such models of learning include Binmore and Samuelson (1993) and Börgers and Sarin (1994), who both explore the relationship with 'replicator dynamics' models, and Gilboa and Schmeidler (1992, 1993) who develop an axiomatic "case-based decision theory" where players satisfice relative to aspiration levels that are based on past experimence.

The structure of interaction between players in our model does not correspond to random matching of pairs selected from a certain population, as in the 'evolutionary' literature – cf., for example, Binmore and Samuelson (1993), Börgers and Sarin (1994), Kandori, Mailath and Rob (1993) or Young (1993). In our context, a given pair of agents plays the game repeatedly over time. This stands in contrast with some of the well-known models whose concern is to provide an evolutionary basis for the rise of cooperation, e.g., Robson (1990), Fudenberg and Maskin (1990), or Binmore and Samuelson (1992) They consider situations where the *repeated* Prisoner's Dilemma is recurrently being played between pairs of individuals randomly selected from the general population. In a sense, their objective is to select among multiple equilibria of the underlying supergame by embedding it in a wider intertemporal framework. Instead, our approach remains within the the scenario of a single indefinite repetition of the stage game, singling out the *stage* outcome which happens to be played most of the time in the long run Bendor, Mookherjee and Ray (1992) is a precursor to this paper. In that paper, aspirations were assumed to be fixed, and models of reinforcement learning that led to long run outcomes consistent with aspiration levels in general two player repeated games were analyzed. Such an equilibrium notion is appropriate for characterizing long run outcomes in contexts of evolving aspirations where the latter converge. This paper extends this model to include an explicit process of evolving aspirations as well as of strategies of players, but in the context of a specific game, the Prisoner's Dilemma. Aspirations do turn out to converge, and the long run outcome is essentially cooperative, thus vindicating the solution concept used in that paper.

Kim (1995) and Pazgal (1995) both apply the Gilboa-Schmeidler case-based theory to games of coordination or the Prisoners Dilemma. They allow aspirations to evolve simultaneously with the strategies selected by players in a context of repeated interaction, and provide conditions under which long run outcomes entail cooperation. These conditions entail initial aspiration levels lying in prespecified ranges: for instance, Pazgal needs to assume that they are sufficiently high relative to the cooperative payoffs for both players, while Kim assumes that they lie slightly below the cooperative payoffs for both players. Our model in contrast predicts cooperation in the long run, irrespective of initial conditions average maximal experienced payoffs in past plays, whereas we assume that aspirations average maximal experienced payoffs in past plays, whereas we assume they average the actual experienced payoffs. Hence their theory imparts a certain additional degree of "ambitiousness" to players, which helps in ensuring convergence to cooperative outcomes.

### 6 Concluding Comments

The learning dynamics studied in this paper is admittedly stylized and the context to which it is applied undoubtedly special. However, we want to conclude the main body of the paper by discussing its robustness in the face of alternative specifications, outlining as well a variety of different possible extensions.

As amply discussed, the essential role of trembles in our framework is that of liberating the process from "fragile" pss's And, as it turns out, only the (C, C) pss proves to be resilient enough to display any significant long-run weight, as the adjustment of aspirations becomes very gradual. In this light, it seems clear that a wide variety of alternative perturbations of the process would tend to produce the same results. Consider, for example, a model where aspirations always adjust deterministically according to (1) but, occasionally (i.e., with some small probability  $\eta$ ) players adjust their actions arbitrarily (that is, *not* according to the function p). This would be a formulation more in line with the approach of recent evolutionary literature (where players are assumed to "experiment") and would seem to lead to results which are qualitatively the same as described here. It would still be true that a transition from the pss (C, C) to any of the other three pps's will be a very unlikely event after one tremble, if  $\lambda$  is close to one. Reciprocally, the converse transitions would still require very little inertia (and, therefore, would be much more likely), after one single tremble. The logic underlying our analysis would seem to be essentially unaffected by this alternative specification.

Another variation on the model pertains to the postulated law of motion on aspirations. The geometric formulation considered here could be replaced with

$$\alpha_{t+1} = \frac{1}{t+1} \sum_{\tau=0}^{t} \pi_{\tau}^{1}, \tag{2}$$

that is, a simple average of past payoffs In a heuristic sense, (2) could be interpreted as a limiting case of our framework when  $\lambda \to 1^{-10}$  Whether this new feature also leads to similar conclusions is an open question. The main issue here is that, in this alternative context, *any* transition becomes progressively more lengthy as time proceeds. This could conceivably lead to non-ergodic behavior despite the fact that the *relative* likelihoods of the different transitions across pss's would display the same features as described in our model.

Let us now outline some interesting extensions of the game studied here We shall focus on three of them: (a) coordination games; (b) random-matching population contexts; (c) richer intertemporal strategies. The extent to which we feel we understand each of these extensions is decreasing in the order mentioned.

First, it is not difficult to see that the same asymmetry between upward and downward movements in aspiration levels would appear in strict coordination games (i.e., symmetric games where every main-diagonal payoff is positive, all others being equal to zero). If the game has a unique equilibrium which dominates all others, it can be shown that the long-run outcome of a process analogous to the one described in this paper would coincide with such an equilibrium.

A second extension would involve a finite-population context where players are randomly matched every period to play a game, say a Prisoner's Dilemma. Even though we have not worked out the details of such a model, we conjecture that the cooperative outcome (i.e., all players choosing C) would still be the unique long-run outcome for  $\lambda$ close to one. The reason is that, in this context, the destabilizing effect of one tremble would still be the same as before, once we are allowed to specify the particular outcome (or chain of outcomes) of the matching mechanism. Since every matching outcome has some positive probability, so has any finite chain of them required for the perturbation to operate in the desired direction. Combining this ideas with an analogous role for

<sup>&</sup>lt;sup>10</sup>Of course, one could not carry out this limit operation literally since this would amount to freezing the aspiration level at its initial value

inertia still present when  $\lambda$  is large, the logic used above would seem applicable to the population context outlined.

Finally, it might be of interest to allow for players who may rely on the intertemporal nature of the interaction and use history-dependent rules. In this context, players could end up using certain rules not only due to their "direct" payoff performance but also, indirectly, because of the behavior they induce in others. Again, one would postulate reinforcing mechanisms of the type described on the set of possible rules, the learning process being occasionally perturbed by stochastic trembles. A preliminary issue which would arise in this respect pertains to the specification of the space of rules to be considered. Initially, one might consider simply addding to the "flat" (history-independent) rules, a simple reactive one (for example, in the Prisoner's Dilemma, a strategy of the Tit-For-Tat variety). In this framework, it may be possible to shed some light on the difficult issue of whether sophisticated rules may either arise or, at least, play a crucial role in facilitating the emergence of simple cooperative behavior.

## 7 Proofs

**Proof of Proposition 1.** Given initial aspirations  $(\alpha_0, \beta_0)$ , aspirations at all later dates will be contained in the convex hull of  $(\alpha_0, \beta_0)$  and the four (pure) pavoff points of the game. Let this convex hull be denoted C and let the maximum aspirations for the two players in C be denoted by  $\bar{\alpha}$  and  $\bar{\beta}$  respectively.

Consider any state  $(\alpha, \beta, A, B)$  with  $(\alpha, \beta) \in C$ . Let the payoffs generated by the action pair (A, B) be denoted  $(\pi^1, \pi^2)$ . Consider the probability of an infinite run on this action pair, which is given by

$$h(\alpha,\beta) \equiv \prod_{t=1}^{\infty} p(\lambda^t(\alpha - \pi^1)) p(\lambda^t(\beta - \pi^2)).$$

We claim that (i)  $h(\alpha,\beta) > 0$  for every  $(\alpha,\beta) \in C$ , and (ii) h is nonincreasing in each argument.

To prove (i), it suffices to check that:

$$\sum_{t=1}^{\infty} [1 - l(\lambda^t(\alpha - \pi^1)), \lambda^t(\beta - \pi^2))] < \infty,$$

where  $l(d_1, d_2)$  denotes  $p(d_1)p(d_2)$ . This condition is satisfied because the left hand side is bounded above by

$$M\frac{\lambda}{1-\lambda}\{|\alpha-\pi^1|+|\beta-\pi^2|\}.$$

thus establishing (i). Claim (ii) follows directly from the fact that p is nonincreasing

Given (i) and (ii), we see that for every  $(\alpha, \beta) \in C$ ,

$$h(\alpha,\beta) \ge h(\bar{\alpha},b) > 0.$$

Let  $\epsilon > 0$  denote the minimum value of  $h(\bar{\alpha}, \bar{\beta})$  across all possible initial action pairs. It follows that at every date the probability of converging to the pure strategy state corresponding to the ongoing action pair is at least  $\epsilon$ , thereby completing the proof.

We now prepare for the statement and proof of Theorem 2. Let  $\Delta(E)$  denote the set of probability measures on a compact state space E, endowed with the Borel  $\sigma$ -algebra. For any transition probability Q on E and any measure  $\mu \in \Delta(E)$ , define  $\mu \cdot Q \equiv \int_E \mu(x)Q(x, \cdot)$ . Two transition probabilities P and Q naturally induce a third PQ through composition:  $PQ(x, \cdot) \equiv P(x, \cdot) \cdot Q$ . This permits us to define *m*-step transition probabilities iteratively:  $P^m(x, \cdot) = P^{m-1}(x, \cdot) \cdot P$ , where  $P^0$  is the degenerate probability on x.

Given a real valued function f on E, define  $Pf \equiv \int_E P(x, dy) f(y)$ .

A measure  $\Pi$  on E is invariant with respect to P if  $\Pi_{-}P = \Pi_{-}$ 

Perturb the transition probability P in the following manner: define for  $\eta \in (0, 1)$ ,

$$P^{\eta} = (1 - \phi(\eta))P + \phi(\eta)Q^{\eta} Q^{\eta} = (1 - \psi(\eta))Q + \psi(\eta)Q^{\eta}_{1}$$

where  $\phi(\eta) \to 0, \psi(\eta) \to 0, 0 < \phi(\eta) < 1, 0 \le \psi(\eta) < 1$  as  $\eta \to 0$ 

THEOREM 2 Assume that

[a] For each  $x \in E$ ,  $\frac{1}{T+1} \sum_{t=0}^{T} P^t(x, \cdot)$  converges weakly to a probability measure  $R(x, \cdot)$ . [b] Q has the strong Feller property: Qf is continuous for all bounded measurable f on E.

[c] Q is open set irreducible, i.e., for all open sets U and all  $x \in E$ ,  $\sum_{n=1}^{\infty} Q^n(x, U) > 0$ ; and

[d] QR has a unique invariant measure  $\Pi^*$ .

Then  $P^{\eta}$  has a unique invariant measure  $\Pi^{\eta}$ , which converges weakly to  $\Pi^*$  as  $\eta \downarrow 0$ .

**Proof.** Given any  $\eta > 0$ , properties [b] and [c] imply that  $P^{\eta}$  is a T-chain Applying Theorem 16.2.5 in Meyn and Tweedie (1993),  $P^{\eta}$  is uniformly ergodic, and has a unique invariant measure  $\Pi^{\eta}$  Then

$$\Pi^{\eta} \left[ (1 - \phi(\eta)) P + \phi(\eta) Q^{\eta} \right] = \Pi^{\eta},$$

implying that

$$\Pi^{\eta}[\phi(\eta)Q^{\eta}] = \Pi^{\eta} - (1 - \phi(\eta))\Pi^{\eta}P.$$

Given any bounded continuous function f, apply the probability measures above to  $P^m f$ :

$$\Pi^{\eta} [\phi(\eta)Q^{\eta}] P^{m} f = \Pi^{\eta} P^{m} f - (1 - \phi(\eta)) \Pi^{\eta} P^{m+1} f$$

Multiplying by  $(1 - \phi(\eta))^t$  and summing over t = 0, ..., T we obtain

$$\Pi^{\eta} (\phi(\eta)Q^{\eta}) \{ \sum_{t=0}^{T} (1-\phi(\eta))^{t} P^{t} f \} = \Pi^{\eta} f - (1-\phi(\eta))^{T+1} \Pi^{\eta} P^{T+1} f$$

Taking  $T \to \infty$ , and using  $\sup_x |\Pi^{\eta} P^{T+1} f(x)| \leq \sup_x |f(x)|$ , we get

$$\Pi^{\eta} \{ Q^{\eta} g^{\eta} \} = \Pi^{\dot{\eta}} f, \tag{3}$$

where

$$g^{\eta}(x) = \phi(\eta) \sum_{t=0}^{\infty} (1 - \phi(\eta))^t P^t f(x).$$

Note that  $\lim_{T\to\infty} \frac{1}{T+1} \sum_{t=0}^{T} P^t f(x) = Rf(x)$  implies  $g^{\eta}(x) \to Rf(x)$  as  $\eta \downarrow 0$ . Now

$$\Pi^{\eta} Q^{\eta} g^{\eta} = (1 - \psi(\eta)) \Pi^{\eta} Qg^{\eta} + \psi(\eta) \Pi^{\eta} Q_1^{\eta} g^{\eta}$$

$$\tag{4}$$

Since  $|g^{\eta}(x)| \leq \sup_{y} |f(y)| \equiv M < \infty$ , we also have  $|\Pi^{\eta} Q_{1}^{\eta} g^{\eta}| \leq M$ . As Q is strong Feller, it follows that  $Qg^{\eta}$  and QRf are continuous. Hence

$$\sup_{x} |Qg^{\eta}(x) - QRf(x)| \to 0$$

as  $\eta \downarrow 0$ . Thus, if  $\eta_n$  is any sequence with  $\eta_n \downarrow 0$  with  $\Pi^{\eta_n} \to \hat{\Pi}$  in the topology of weak convergence,

$$\Pi^{\eta_n} Qg^{\eta_n} \to \hat{\Pi} QRf.$$
(5)

(4) and (5) together imply that

$$\lim_{n} \Pi^{\eta_n} \cdot Q^{\eta_n} g^{\eta_n} = \hat{\Pi} \cdot QRf \tag{6}$$

\*

Combining (3) and (6),

$$\hat{\Pi}.QRf = \hat{\Pi}f.$$

It then follows from [d] that  $\hat{\Pi} = \Pi^*$ , which completes the proof **Proof of Propositions 3 and 4.** Define

$$\phi(\eta) \equiv \eta^2 + 2\eta(1-\eta),$$
  
$$\psi(\eta) \equiv \frac{\eta^2}{\eta^2 + 2\eta(1-\eta)}.$$

It is then evident that

$$P^{\eta} = [1 - \phi(\eta)]P + \phi(\eta)Q^{\eta}$$

where

$$Q^{\eta} \equiv [1 - \psi(\eta)]Q + \psi(\eta)Q^*,$$

 $Q^{\ast}$  denoting the transition probability when both players are subjected simultaneously to a tremble.

Assumption (a) of Theorem 2 is established by Proposition 1, while assumptions (b) and (c) are valid by construction. Hence it remains to check assumption (d), i.e., that the process QR has a unique invariant distribution. We know from Proposition 1 that every invariant distribution of QR must be concentrated on the four pss's. Hence it suffices to show that there exists a pss (the mutual defection pss) which can be reached with positive probability from every other pss in the QR process.

For this it suffices to check that the action pair (D, D) will be played at some date, when we start at any pss and subject the aspiration of one player to a tremble. The reason is that once this happens, the reasoning of Proposition 1 implies that with positive probability there will be an infinite run on (D, D) thereafter, causing convergence to the (D, D) pss. If we start with the (D, D) pss then this is obvious, owing to inertia. And if we start with any other pss, then one small upward tremble to one player's aspiration will cause (C, D) or (D, C) followed by (D, D) to be played with some probability.

For the proof of Theorem 1, we need some additional notation. If  $\pi_s$  denotes the pavoff to player 1 in any period s, then

$$\alpha_{s+1} = \lambda \alpha_s + (1 - \lambda) \pi_s,$$

so that

$$|\alpha_s - \alpha_{s+1}| = (1 - \lambda)|\alpha_s - \pi_s| \le (1 - \lambda)W,$$

where W is the maximum conceivable value between aspirations and payoffs (the width of the compact interval  $\Lambda$  to which aspirations belong).

Let T be the minimum number of periods that need to elapse before aspirations at time T + 2 are different from aspirations at time 0 by an amount not less than a. Then it is clear that  $T \ge T(\lambda, a)$ , where  $T(\lambda, a)$  is the smallest integer such that

$$(1-\lambda)[T(\lambda,a)+2]W \ge a.$$

It follows that

$$(1-\lambda)T(\lambda,a) \ge \frac{a}{W} - 2(1-\lambda).$$
 (7)

Note that if a > 0, then  $T(\lambda, a) > 0$  for all  $\lambda$  sufficiently close to unity, and indeed, that  $T(\lambda, a) \to \infty$  as  $\lambda \to 1$ . This construct will be used at various stages below.

For any  $(\hat{a}, \hat{b}) \gg 0$  such that  $\delta + \hat{a} < \sigma - \hat{a}$  and  $\delta + \hat{b} < \sigma - \hat{b}$ , let  $I(\hat{a}, \hat{b})$  be the rectangle defined by  $[\delta + \hat{a}, \sigma - \hat{a}] \times [\delta + \hat{b}, \sigma - \hat{b}]$ 

LEMMA 1 Consider any  $(\hat{a}, \hat{b}) \gg 0$  such that  $\delta + \hat{a} < \sigma - \hat{a}$  and  $\delta + \hat{b} < \sigma - \hat{b}$ . Then given any  $\epsilon > 0$ , there exists  $\lambda_1 \in (0, 1)$  such that

$$\operatorname{Prob}^{\lambda}((\alpha_t,\beta_t,s_t)\to(\sigma,\sigma,(C,C))|(\alpha_T,\beta_T)\in I(\hat{a},\hat{b}))>1-\epsilon$$

for all T and all  $\lambda \in (\lambda_1, 1)$ .

**Proof.** Fix any T with  $(\alpha_T, \beta_T) \in I(\hat{a}, \hat{b})$ . Let  $T^*(\lambda)$  be the minimum number of periods t such that  $(\alpha_{T+t+2}, \beta_{T+t+2}) \notin I(\frac{\hat{a}}{2}, \frac{\hat{b}}{2})$ . Observe that if  $a \equiv \frac{1}{2} \min\{\hat{a}, \hat{b}\}$ , then that  $T^*(\lambda) \geq T(\lambda, a)$  (see (7)). It follows that a lower bound for  $T^*(\lambda)$  can be found that is independent of T.

We observe, next, that there is  $\zeta > 0$  (independent of T and  $\lambda$ ) such that as long as  $t = T + 2, T + 3, \dots, T(\lambda, a) + T$ ,  $s_t = (C, C)$  with probability at least  $\zeta$ . To see this, suppose first that (C, C) is played in period t - 2. In that case (C, C) is played in period t with probability one. If (D, D) is played in period t - 2, then the probability is easily seen to be at least  $p(\frac{\hat{a}}{2})p(\frac{\hat{b}}{2})$ . If at date t - 2, (D, C) is played, then the conditional probability of playing (D, D) at t - 1 is at least  $[1 - p(\delta)]$  (because player 1 will stick for sure to D, and player 2 will switch with probability at least  $[1 - p(\delta)]$ ). Thereafter, a switch to (C, C) in period t occurs with probability at least  $p(\frac{\hat{a}}{2})p(\frac{\hat{b}}{2})$ . The conditional probability in this case is therefore at least  $[1 - p(\delta)]p(\frac{\hat{a}}{2})p(\frac{\hat{b}}{2})$ . Finally, the argument for (C, D) is symmetric.

Thus in all cases, the conditional probability of playing (C, C) at date t is bounded below by the positive number  $\zeta \equiv [1 - p(\delta)]p(\frac{\hat{a}}{2})p(\frac{\hat{b}}{2})$ .

We may now compute a lower bound on the probability of (C, C) being played at least once in  $T(\lambda, a) - 2$  periods. The probability that (C, C) will *never* be played during this stretch is clearly bounded above by  $(1 - \zeta)^{T(\lambda, a) - 2}$ . Thus the required lower bound is given by  $1 - (1 - \zeta)^{T(\lambda, a) - 2}$ . Choose  $\lambda_1$  such that

$$1 - (1 - \zeta)^{T(\lambda_1, a) - 2} \ge 1 - \epsilon$$

We complete the proof by showing that once (C, C) is played during these periods,  $(\alpha_t, \beta_t, s_t) \to (\sigma, \sigma, (C, C))$  as  $t \to \infty$ . To see this, simply observe that whenever (C, C)is played during this time, say at date s,  $(\alpha_s, \beta_s) \leq (\sigma - \frac{b}{2}, \sigma - \frac{b}{2})$  by the construction of  $T(\lambda, a)$  It follows that once (C, C) is played, it will be played repeatedly thereafter, and we are done.

LEMMA 2 Let  $T(\lambda)$  be a sequence of positive integers such that  $T(\lambda) \to \infty$  as  $\lambda \to 1$ For each  $\lambda$ , let  $\{X_t^{\lambda}, Z_t^{\lambda}\}$  be a finite horizon stochastic process with terminal date  $T(\lambda)$ , such that  $X_t$  takes values only in  $\{0, 1\}$  (there is no restriction on  $Z_t$ ) Suppose that for each  $\lambda$ ,  $X_0^{\lambda}$  is equal to a constant  $i(\lambda)$  (which can take the values 0 or 1). Use the notation  $h_t$  to denote t-histories for each  $t \geq 1$ , and the notation  $L(h_t)$  to denote the value of  $X_{t-1}^{\lambda}$  for every t-history.

Suppose that for every  $\lambda \in (0, 1)$ ,

$$\operatorname{Prob}(X_t^{\lambda} = 1 | h_t) \le u < 1,$$

whenever  $L(h_t) = 0$ , while

$$\operatorname{Prob}(X_t^{\lambda} = 1 | h_t) \le v < 1,$$

if  $L(h_t) = 1$ . Then for every  $\nu > 0$ ,

$$\operatorname{Prob}\left\{\frac{1}{T(\lambda)+1}\sum_{t=0}^{T(\lambda)}X_t^{\lambda} \le \frac{u+\nu}{1-v+u}\right\} \to 1 \text{ as } \lambda \to 1.$$
(8)

**Proof.** We begin with a coupling result that will be needed for the proof.

CLAIM. Suppose  $\{U_0, U_1, U_2, \ldots, U_T\}$  is a finite sequence of random variables that assume values in  $\{0, 1\}$ . Suppose that  $U_0 = i$  (where *i* is either 0 or 1), and that for  $t \ge 1$ ,

$$\operatorname{Prob}(U_t = 1|h_t) = p_t(h_t) \le u < 1,$$

if  $L(h_t) = 0$ , and

$$\operatorname{Prob}(U_t = 1|h_t) = p_t(h_t) \le v < 1,$$

if  $L(h_t) = 1$ .

Let  $\{V_1, V_2, \ldots, V_T\}$  be a (finite-horizon) Markov Chain with values in 0, 1 such that  $V_0 = i$  as well and for all  $t \ge 0$ 

$$Prob(V_{t+1} = 1 | V_t = 0) = u \quad Prob(V_{t+1} = 0 | V_t = 0) = 1 - u$$
$$Prob(V_{t+1} = 1 | V_t = 1) = v \quad Prob(V_{t+1} = 0 | V_t = 0) = 1 - v$$

Then for all x,

$$Prob(U_1 + U_2 + \dots + U_T \le x) \ge Prob(V_1 + V_2 + \dots + V_T \le x)$$
(9)

We prove this using a standard coupling argument. Define q(0) = u and q(1) = vTake a probability space  $\Omega = [0, 1]^T$  with the Borel  $\sigma$ -field on  $\Omega$ , and product Lebesgue measure.

For  $x = (x_1, x_2, \ldots, x_T) \in \Omega$ , let  $\tilde{U}_0(x) = i$ . Recursively, for  $t \ge 1$ , having defined  $\tilde{U}_0, \ldots, \tilde{U}_{t-1}$ ;  $\tilde{V}_1, \ldots, \tilde{V}_{t-1}$ , let  $\xi_t(x) \equiv p_t(\tilde{U}_0(x), \tilde{U}_2(x), \ldots, \tilde{U}_{t-1}(x))$  and  $\eta_t(x) = q(\tilde{V}_{t-1}(x))$ . Now define  $\tilde{U}_t, \tilde{V}_t$  by

$$\tilde{U}_t(x) = \mathbb{1}_{\{x_t \leq \xi_t(x)\}}$$

and

$$V_t(x) = 1_{\{x_t \le \eta_t(x)\}^+}$$

Observe now that for all t = 0, 1, ..., T, and each  $x \in \Omega$ ,

$$\dot{U}_t(x) \le \dot{V}_t(x). \tag{10}$$

To see this, proceed inductively, noting that the statement is trivially true at t = 0. Recursively, if the statement is true at some t - 1, note that by the construction of U, V and q,

$$\xi_t(x) \leq \eta_t(x).$$

The required inequality (10) at date t follows right away from this observation.

Finally, observe that using product Lebesgue measure, the distribution of  $(U_0, U_1, \ldots, U_T)$  is the same as the distribution of  $(U_0, U_1, \ldots, U_T)$ , and also that the distribution of  $(\tilde{V}_0, \tilde{V}_1, \ldots, \tilde{V}_T)$  is the same as the distribution of  $(V_0, V_1, \ldots, V_T)$ .

Now the required conclusion (9) follows immediately, because the corresponding statement for  $(\tilde{U}_0, \ldots, \tilde{U}_T)$ ,  $(\tilde{V}_0, \ldots, \tilde{V}_T)$  is a direct consequence of (10).

With the Claim in hand, we return to the proof of the lemma. It will be sufficient to establish the lemma for the case where  $i(\lambda)$  is a constant *i*, independent of  $\lambda$ . The general case then follows easily from a subsequence argument.

Let  $\{V_t\}$  be the Markov chain constructed in the Claim. Then interpreting  $X_t^{\lambda}$  as  $U_t$ , (9) implies immediately that

$$\operatorname{Prob}\left(\frac{1}{T(\lambda)+1}\sum_{t=0}^{T(\lambda)}X_t^{\lambda} \le \frac{u+\nu}{1-\nu+u}\right) \ge \operatorname{Prob}\left(\frac{1}{T(\lambda)+1}\sum_{t=0}^{T(\lambda)}V_t \le \frac{u+\nu}{1-\nu+u}\right).$$

By the strong law of large numbers for Markov chains, the RHS above converges to 1, and we are done.

LEMMA 3 There exist positive numbers a and b with  $\delta + b < \sigma - b$ , with the property that for any  $\epsilon > 0$ , there exists  $\lambda_2 \in (0, 1)$  such that

$$\operatorname{Prob}\{(\alpha_t, \beta_t, s_t) \to (\sigma, \sigma, (C, C)) | (\alpha_T, \beta_T) \in N\} \ge 1 - \epsilon \tag{11}$$

for every T and  $\lambda \in (\lambda_2, 1)$ , where  $N \equiv [\delta - a, \delta + a] \times [\delta + 2b, \sigma - 2b]$ 

**Proof.** We begin the proof by choosing the numbers a and b, and defining some other values that will be needed in the argument.

Start by choosing b > 0 such that  $0 < \delta - 2b < \delta + 2b < \sigma - 2b$ , and define the rectangle  $M_1$  by

$$M_1 \equiv \{(\alpha, \beta) | \alpha \in [\delta - b, \delta + b], \beta \in [\delta + 2b, \sigma - 2b] \}.$$

Consider the process commencing from some aspiration vector  $(\alpha_T, \beta_T) \in M_1$  at date T, and any given vector of actions. For each  $\lambda$ , note that a lower bound on the the minimum number of periods after which  $(\alpha_{T+t+2}, \beta_{T+t+2})$  fails to lie within the larger rectangle,  $M_2$ , defined by

$$M_2 \equiv \{(\alpha, \beta) | \alpha \in [\delta - 2b, \delta + 2b], \beta \in [\delta + b, \sigma - b] \}$$

is given by  $T(\lambda, b)$  (see (7)).

Clearly, there is an interval  $[\bar{\lambda}, 1]$  such that for every  $\lambda$  in this interval,  $T(\lambda, b) > 0$ . For the rest of the argument,  $\lambda$  will be taken to lie in this interval.

Define  $K \equiv \frac{b}{W} - 2(1 - \overline{\lambda}) > 0$  It follows then that for all  $\lambda \in (\overline{\lambda}, 1)$ ,

$$(1-\lambda)T(\lambda,b) \ge K > 0. \tag{12}$$

Next, fix  $a \in (0, b]$  and define the following quantities:

$$d' \equiv 1 - p(a) > 0$$
  

$$e' \equiv p(2b)[1 - p(b)] > 0$$
  

$$i' \equiv p(\delta - 2b) < 1.$$
(13)

Observe that the inequalities above hold because  $p(x) \in (0,1)$  whenever x > 0

We will impose a further restriction on the choice of a. It must satisfy the additional restriction that

$$\left[\theta - (\delta + a)\right] \left(e'\chi - \frac{d'}{1 - i' + d'}\right) - a(1 - e'\chi) - \frac{(\delta + a)d'}{1 - i' + d'} > \frac{2a}{K}$$
(14)

where K is given by (12), and  $\chi$  is a strictly positive number satisfying the equality

$$\chi = \min\{p(2b)p(\sigma - \delta - b), p(\delta + b)\} - \hat{\epsilon}$$
(15)

for some small but positive  $\hat{\epsilon}$ 

Let us check to see that this can be done Certainly  $\chi$  can be chosen as in (15) Having done so, note that the RHS of (14) goes to 0 as  $a \to 0$ , while the LHS of (14) converges to the positive quantity

$$[\theta - \delta]e'\chi$$

(to see this, use (13)). By the continuity of both sides of (14) in a at a = 0, it follows that there exists a small but strictly positive such that both  $\delta + a < \sigma - a$  and (17) holds

We complete our construction by noting that there exists  $\epsilon' > 0$  such that if we define

$$d \equiv 1 - p(a) + \epsilon'$$

$$e \equiv p(2b)[1 - p(b)] - \epsilon'$$

$$i \equiv p(\delta - 2b) - \epsilon',$$
(16)

then d, e and i all lie strictly between 0 and 1, and moreover,

$$\left[\theta - (\delta + a)\right] \left(e\chi - \frac{d}{1 - i + d}\right) - a(1 - e\chi) - \frac{(\delta + a)d}{1 - i + d} \ge \frac{2a}{K}$$
(17)

In what follows,  $a, b, d, e, i, \hat{\epsilon}$ , and  $\epsilon'$  are fixed by these considerations, irrespective of the value of  $\lambda \in [\bar{\lambda}, 1)$ .

Let  $W_1 \equiv \{(C, C) \text{ is played for some } T \leq t \leq T(\lambda, b) + T, \text{ and } \alpha_k \leq \delta + a \text{ for } T \leq k \leq t\}$  Note by the construction of  $T(\lambda, b)$  that  $(\alpha_t, \beta_t) \leq (\sigma - b, \sigma - b)$  for all  $t \in \{0, 1, \ldots, T(\lambda)\}$  Consequently, if  $W_1$  occurs, (C, C) will be played repeatedly thereafter, and  $(\alpha_t, \beta_t, s_t) \to (\sigma, \sigma, (C, C))$  for sure. Thus

$$\operatorname{Prob}^{\lambda}\{(\alpha_t, \beta_t, s_t) \to (\sigma, \sigma, (C, C)) | W_1\} = 1.$$
(18)

In what follows, we suppose, then, that the event  $W_1$  does not occur.

Denote by  $S^{\lambda}$  the event  $\{(\alpha_t, \beta_t) \in I(a, b) \text{ for some } T \leq t \leq T(\lambda, b) + T\}$ 

CLAIM. There exists a function  $g(\lambda)$  on  $(\overline{\lambda}, 1)$ , independent of  $T_{\pm}$  with  $g(\lambda) \to 1$  as  $\lambda \to 1$ , such that

$$\operatorname{Prob}^{\lambda}(S^{\lambda}|(\alpha_{T},\beta_{T})\in N,\sim W_{1})\geq g(\lambda)$$
(19)

To establish this claim, let  $W_2$  be the event  $\{\alpha_t \geq \delta + a \text{ for some } t = T, T + 1, \dots, T(\lambda, b) + T - 1\}$ . Note that by the definition of conditional probabilities,

$$\begin{aligned} \operatorname{Prob}^{\lambda}(S^{\lambda}|(\alpha_{T},\beta_{T})\in N,\sim W_{1}) &= \operatorname{Prob}^{\lambda}(S^{\lambda}|(\alpha_{T},\beta_{t})\in N, W_{2},\sim W_{1})\operatorname{Prob}^{\lambda}(W_{2}) \\ &+ \operatorname{Prob}^{\lambda}(S^{\lambda}|(\alpha_{T},\beta_{T})\in N,\sim W_{2},\sim W_{1})\operatorname{Prob}^{\lambda}(\sim W_{2}). \end{aligned}$$

First evaluate  $\operatorname{Prob}^{\lambda}(S^{\lambda}|(\alpha_{T},\beta_{t}) \in N, W_{2}, \sim W_{1})$ . Note that if  $W_{2}$  occurs (at some date t), then by a property of  $T(\lambda, b)$ ,  $\alpha_{t} \leq \delta + 2b \leq \sigma - 2b \leq \sigma - a$ , by the choices of a and b. So  $\alpha_{t} \in [\delta + a, \sigma - a]$ . Also, by construction of  $T(\lambda, b)$ , it must be the case that  $\beta_{t} \in [\delta + b, \sigma - b]$ , so that  $(\alpha_{t}, \beta_{t}) \in I(a, b)$ . This shows that

$$\operatorname{Prob}^{\lambda}(S^{\lambda}|(\alpha_{T},\beta_{T})\in N, W_{2}, \sim W_{1}) = 1.$$

$$(20)$$

It remains to evaluate the conditional probability

$$\operatorname{Prob}^{\lambda}(S^{\lambda}|(\alpha_T,\beta_T)\in N, \sim W_2, \sim W_1)$$

for each T

It will be useful in what follows to concentrate on the action plays in each of the periods  $T, T + 1, \ldots, T(\lambda, b) + T$ . One way of doing this is to note that the overall stochastic process, conditional on some initial action  $s_T$ ,  $(\alpha_T, \beta_T) \in N$  and the event  $\sim W$ , defines a stochastic process (which in general will be non-Markovian) on the

actions  $s_t$  played in periods  $t = T, T + 1, ..., T(\lambda, b) + T$ . This process can be described, given the initial conditions, by a sequence of functions (one for each date), describing the probability of each action pair at date t conditional on the entire history of actions  $h_t$  up to that date. At t, let  $P^{\lambda}(., h_t)$  denote this function. With slight abuse of notation, we will use  $P^{\lambda}$  to denote the probability of various events as well, conditional or otherwise.

Let  $\mu$  denote the fraction of occurrences of (D, D), and  $\gamma$  the fraction of occurrences of (C, D), for the dates  $T, T+1, \ldots, T(\lambda, b)+T$ . Of course,  $\mu$  and  $\gamma$  are random variables for each  $\lambda$ 

Recalling the definitions in (16) and the definition of  $\chi$  in (15), consider the event Z described by

$$\mu \le 1 - e\chi \tag{21}$$

and

$$\gamma \le \frac{d}{1 - i + d} \tag{22}$$

SUBCLAIM. There exists a function  $g(\lambda)$  on  $(\bar{\lambda}, 1)$ , independent of  $T_{+}$  with  $g(\lambda) \to 1$  as  $\lambda \to 1$  such that

$$\operatorname{Prob}^{\lambda}(Z|(\alpha_T,\beta_T)\in N, \sim W_2, \sim W_1) \ge g(\lambda).$$

$$(23)$$

The argument up to the third paragraph following (27) is concerned with the proof of this Subclaim.

Begin by computing  $P^{\lambda}((D, D), h_t)$  for each  $\lambda, t \geq 1$  and  $h_t$ .

Let  $s(h_t)$  denote the last action vector, i.e., the action at date t - 1, under the *t*-history  $h_t$ . If  $s(h_t) = (D, D)$ , then the probability that both players continue with D is at least  $p(2b)p(\sigma - \delta - b)$  (since player 1's aspiration cannot exceed  $\delta + 2b$  and player 2's aspiration cannot exceed  $\sigma - b$ , by the construction of  $T(\lambda, b)$ ). If  $s(h_t) = (C, D)$ , then by a similar argument, the probability of moving to (D, D) the next period is at least  $p(\delta - 2b)$ . If  $s(h_t) = (D, C)$ , the probability of moving to (D, D) is at least  $p(\delta + b)$ . By invoking (15), we see, therefore, that

$$P^{\lambda}((D,D),h_t) \ge \chi + \hat{\epsilon}$$

for all  $h_t$  and all  $\lambda$ .

Let  $E^{\lambda}$  be the event

$$\left\{\frac{1}{T(\lambda,b)+1}\sum_{t=T}^{T(\lambda,b)+T}\mathbf{1}_{(D,D)} \geq \chi\right\},\$$

where the notation  $1_s$  denotes the indicator function of the action vector s. Note that  $\chi$  is independent of T. Now apply Lemma 2, with  $X_t = 0$  whenever (D, D) is played,  $X_t = 1$  if anything else is played, with  $Z_t$  set equal to some constant, with  $u = v = 1 - \chi$ ,

and  $\nu = \hat{\epsilon}$ . We may deduce that there exists a function  $g_1(\lambda)$  (with  $g_1(\lambda) \to 1$  as  $\lambda \to 1$ ) such that

[I]  $P^{\lambda}(E^{\lambda}) \ge g_1(\lambda)$ .

Next, consider the probability  $P^{\lambda}((D,C),h_t)$  for histories such that  $s(h_t) = (D,D)$ . For (D,C) to follow (D,D), player 1 must stay at D while player 2 switches. Because  $\alpha_t \leq \delta + 2b$ , player 1 stays with probability at least p(2b), while because  $\beta_t \geq \delta + b$ , player 2 switches with probability at least [1-p(b)]. Consequently, recalling (16), we see that

$$P^{\lambda}((D,C),h_t) \ge e + \epsilon' \tag{24}$$

for all  $t \geq 1$ , all  $\lambda$ , and all  $h_t$  with  $s(h_t) = (D, D)$ 

Now, consider the event

$$F^{\lambda} \equiv \{\sum_{t=T}^{T(\lambda,b)+T} 1_{(D,C)} \ge e\chi T(\lambda,b)\},\$$

where  $\chi$  is the constant used to define  $E^{\lambda}$ , and a particular conditional probability  $P^{\lambda}(F^{\lambda}|E^{\lambda})$ . Because the event  $E^{\lambda}$  occurs in the conditioning, the number of occurrences of (D, D) is at least as big as  $\chi T(\lambda, b)$  for each  $\lambda$ .

Use this information to construct a stochastic process as follows. Each occurrence of (D, D) is to be treated as a "date". The number of dates will be taken to be the greatest integer not exceeding  $\chi T(\lambda, b)$ : this is  $T(\lambda)$  in Lemma 2. Throw away all information after this date. Let  $X_t$  be the following random variable that describes the action vector immediately following the *t*th realization of (D, D): X = 0 if (D, C) occurs, and X = 1 otherwise. Let  $Z_t$  be a list of all the action vectors that follow the *t*th occurrence of (D, D), up to the t + 1 th occurrence of (D, D). [If (D, D) is immediately followed by another (D, D), then set Z equal to some arbitrary constant.] This process fits all the conditions of Lemma 2, if both u and v are identified with  $1 - e - \epsilon'$ . Applying the lemma, we may conclude that there exists a function  $g_2(\lambda)$  with  $g_2(\lambda) - 1$  as  $\lambda \to 1$  such that

$$[\mathbf{II}] P^{\lambda}(F^{\lambda}|E^{\lambda}) \ge g_2(\lambda)$$

Again, note that  $g_2$  is independent of T, since e is.

Next, consider the probability  $P^{\lambda}((C, D), h_t)$  First suppose that  $s(h_t) = (D, D)$ . For (C, D) to follow (D, D), player 1 must switch to C while player 2 stays at D. Because we are conditioning on  $\sim W$  and so in particular on  $\sim W_2$ , we have  $\alpha_t \leq \delta + a$ . It follows that the transition occurs with probability no more than [1 - p(a)] It follows (recall (16)) that

$$P^{\lambda}((C,D),h_t) \le 1 - p(a) = d - \epsilon'$$
(25)

for all  $t \ge 1$ , all  $\lambda$ , and all  $h_t$  with  $s(h_t) = (D, D)$ 

Next, suppose that  $s(h_t) \neq (D, D)$ . In this case,  $P^{\lambda}((C, D), h_t)$  can only be positive when (C, D) itself was played last (remember that we are conditioning on  $\sim W$ ). This requires that player 1 stick to his previous action, which will occur with probability at most  $p(\delta - 2b)$ . Using (16), it follows that

$$P^{\lambda}((C,D),h_t) < i + \epsilon' \tag{26}$$

for all  $t \ge 1$ , all  $\lambda$ , and all  $h_t$  with  $L(h_t) \ne (D, D)$ .

Consider, then, the event

$$G^{\lambda} \equiv \left\{ \sum_{t=T}^{T(\lambda)+T} \mathbb{1}_{(C,D)} \le \frac{d}{1-i+d} \right\}$$

We claim that there exists a function  $g_3(\lambda)$ , independent of T, with  $g_4(\lambda) \to 1$  as  $\lambda \to 1$  such that

[III]  $P^{\lambda}(G^{\lambda}) \geq g_3(\lambda)$ .

Proving this claim requires the application of Lemma 2 yet again Start with the (unconditional) event  $G^{\lambda}$ . Define a stochastic process as in Lemma 2 with  $T(\lambda) = T(\lambda, b)$ , with  $X_t = 1$  if the action vector at time t+T is (C, D) (and 0 otherwise), with  $u = d - \epsilon'$  and  $v = i + \epsilon'$ , and with  $\nu = \epsilon'$ . Take  $Z_t$  to be some constant for all t. Now the lemma applies, so we see that there exists a function  $g_3(\lambda)$  with the required properties.

We may now combine observations [I]-[III]. The point is to recognize that all these three hold, then it must be the case that

$$P^{\lambda}(G^{\lambda} \cap F^{\lambda} \cap E^{\lambda}) \ge g(\lambda), \tag{27}$$

for some function  $g(\lambda)$  that is independent of T and with the property that  $g(\lambda) \to 1$  as  $\lambda \to 1$ .

To complete the proof of (23), we unravel what the event  $G^{\lambda} \cap F^{\lambda} \cap E^{\lambda}$  implies for the values of  $\mu$  and  $\gamma$ , which, it will be recalled, are the fractions of (D, D)'s and (C, D)'s respectively during the dates  $T, T + 1, \ldots, T(\lambda, b) + T$ .

Let  $\kappa$  denote the fraction of (D, C)'s during this period. Note, first, that  $\mu + \kappa \leq 1$ , while under the event  $F^{\lambda}$ ,  $\kappa \geq e\chi$ . Combining these two observations, we may conclude that  $\mu \leq 1 - e\chi$ . This shows that (21) must hold under the events  $E^{\lambda}$  and  $F^{\lambda}$ .

Next, note that under the event  $G^{\lambda}$ ,  $\gamma \leq \frac{d}{1-i+d}$ , which is, of course, (22).

The observations in the last two paragraphs, coupled with (27), establish (23), and the proof of the Subclaim is complete.

Suppose, then, that the conditional event described by (23) does in fact occur Let us find a lower bound on the change in player 1's aspirations as a result of this event Recall that  $\alpha_{t+1} = \lambda \alpha_t + (1 - \lambda)\pi_t$ , where  $\pi_t$  is the payoff at date t, so that

$$\alpha_{t+1} - \alpha_t = (1 - \lambda)(\pi_t - \alpha_t) \tag{28}$$

Recalling that  $\alpha_t \leq \delta + a$  for  $t = 0, 1, \dots, T(\lambda, b) - 1$  (i.e., the event  $W_2$  does not occur), we see that the RHS of (28) is bounded below by  $(1 - \lambda)[\theta - (\delta - a)]$  when the action (D, C) is played, by  $-(1 - \lambda)a$  when the action (D, D) is played, and by  $-(1 - \lambda)(\delta + a)$ when the action (C, D) is played. Using (21) and (22), we may therefore compute the total rightward drift over  $T(\lambda, b)$  periods as

$$T(\lambda,b)(1-\lambda)\{[\theta-(\delta-a)][1-\gamma-\mu]-\mu a-\gamma(\delta+a)\}$$

$$\geq T(\lambda,b)(1-\lambda)\{[\theta-(\delta-a)]\left(e\chi-\frac{d}{1-i+d}\right)-a(1-e\chi)-\frac{(\delta+a)d}{1-i}\}$$

$$\geq T(\lambda,b)(1-\lambda)\frac{2a}{K}\geq 2a,$$

using (12) and (17). We conclude, then, that under the event Z,

$$\alpha_{T(\lambda,b)} \ge \delta + a,\tag{29}$$

while, by the construction of  $T(\lambda, b)$ ,

$$\beta_{T(\lambda)} \ge \delta + b. \tag{30}$$

From (29) and (30), it follows right away that

$$\operatorname{Prob}^{\lambda}(S^{\lambda}|(\alpha_{T},\beta_{T})\in N, \sim W_{2}, \sim W_{1}) \geq g(\lambda), \qquad (31)$$

where  $g(\lambda)$  was introduced in (27).

By combining (20) and (31), and defining  $g(\lambda) \equiv \min\{\hat{g}(\lambda), \tilde{g}(\lambda)\}$ , we obtain (19). Recalling (19), we may conclude that

$$\operatorname{Prob}^{\lambda}(S^{\lambda}|(\alpha_{T},\beta_{T})\in N,\sim W_{1})\geq g(\lambda)$$
(32)

for some function  $g(\lambda)$ , independent of T, such that  $g(\lambda) \to 1$  as  $\lambda \to 1$ . This completes the proof of the Claim.

To complete the proof of the lemma, observe, noting carefully the definition of  $W_{1,\varepsilon}$  that

$$\operatorname{Prob}^{\lambda}\{(\alpha_{t},\beta_{t},s_{t}) \to (\sigma,\sigma,(C,C)) | (\alpha_{T},\beta_{T}) \in N, \sim W_{1}\}$$

$$\geq \sum_{S=T}^{T(\lambda,b)+T} \operatorname{Prob}^{\lambda}\{(\alpha_{t},\beta_{t},s_{t}) \to (\sigma,\sigma,(C,C)) | (\alpha_{S},\beta_{S}) \in I(a,b)\}$$

$$= \operatorname{Prob}^{\lambda}((\alpha_{S},\beta_{S}) \in I(a,b) | (\alpha_{T},\beta_{T}) \in N, \sim W_{1}\}$$
(33)

Using Lemma 1, we may conclude that there exists a function  $g'(\lambda)$  independent of S, with  $g'(\lambda) \to 1$  as  $\lambda \to 1$ , such that

$$\operatorname{Prob}^{\lambda}\{(\alpha_{t},\beta_{t},s_{t})\to(\sigma,\sigma,(C,C))|(\alpha_{S},\beta_{S})\in I(a,b)\}\geq g'(\lambda)$$

Using this observation in (33), we see that

$$\operatorname{Prob}^{\lambda}\{(\alpha_{t},\beta_{t},s_{t}) \to (\sigma,\sigma,(C,C)) | (\alpha_{T},\beta_{T}) \in N, \sim W_{1}\} \\ \geq g'(\lambda) \sum_{S=T}^{T(\lambda,b)+T} \operatorname{Prob}^{\lambda}((\alpha_{S},\beta_{S}) \in I(a,b) | (\alpha_{T},\beta_{T}) \in N, \sim W_{1}\},$$

and now using (32),

$$\operatorname{Prob}^{\lambda}\{(\alpha_t, \beta_t, s_t) \to (\sigma, \sigma, (C, C)) | (\alpha_T, \beta_T) \in N, \sim W_1\} \ge g'(\lambda)g(\lambda), \tag{34}$$

the RHS of which converges to one as  $\lambda \to 1$ , uniformly in T. Combining (18) and (34), the proof of the lemma is complete.

LEMMA 4 Let a and b be positive numbers satisfying  $a < \min\{\delta, b\}$ . For any  $\epsilon > 0$  there exists  $\lambda_3 \in (0, 1)$  such that:

$$\operatorname{Prob}^{\lambda}[\alpha_T < a | J_t] < \epsilon. \tag{35}$$

for any  $\lambda \in (\lambda_3, 1)$ , any event  $J_t$  which is a subset of the event that  $\{\alpha_t > b, \beta_t \leq \theta\}$ , and any pair of dates t, T satisfying T > t.

**Proof.** Suppose that the conditional event described in (35) occurs. Pick  $\lambda_3$  to satisfy  $[p(a)]^{T(\lambda_3,b^*-a)} < \epsilon$ , where  $b^* \equiv \min\{b,\delta\} > a$ . Take any  $\lambda \in (\lambda_3,1)$  and any t, T > t. Define intervening dates l and m as follows: l is the *last* date k between t and T such that  $\alpha_k \geq b^*$ , and m is the *first* date k when  $\alpha_k < a$ . Clearly,  $m - l \geq T(\lambda, b^* - a)$ , and  $a < \alpha_k < b^*$  for all intervening dates. Note also that if  $\beta_t \leq \theta$  then  $\beta_k \leq \theta$  for all  $k \geq t$ , since  $\theta$  is the highest attainable payoff in the game.

We claim that the action pair (C, D) must have been played successively between land m-1. First, note that (C, D) must have been played at date l: any other action pair could not have lowered player 1's aspiration below  $b^*$  at l+1. Now  $\alpha_{l+1} < \delta$ , while  $\beta_{l+1} \leq \theta$ : so player 2 is satisfied at  $t_1$ , while 1 is disappointed. Hence only player 1 has an incentive to switch: (C, D) or (D, D) must be played at l+1. If player 1 does switch to a D at l+1, [s]he will obtain a payoff of  $\delta$ , greater than his/her aspiration. Player 2 may or may not switch thereafter, but irrespective of this, player 1 will obtain a payoff of at least  $\delta$ , and so must stick with D at least until m, since his/her aspiration is less than  $\delta$  until then. Moreover, her aspiration must go up successively from l+1 onwards, so her aspiration cannot fall below a at m. Hence (C, D) must be repeated at l+1. The same argument can then be applied successively to all dates  $l+2, \ldots, m-1$ .

Hence player 1 must have persisted with C between dates l and m despite being disappointed by at least a at every date. The probability of this event is therefore at most  $[p(a)]^{m-l} \leq \epsilon$ . Since this bound is independent of l and m, the lemma follows.

LEMMA 5 For any  $\epsilon^* > 0$ , there exists  $\lambda_4 \in (0, 1)$  such that

$$\operatorname{Prob}\{(\alpha_t, \beta_t, s_t) \to (\delta, \delta, (D, D)) | (\alpha_0, \beta_0) \ge (\sigma, \sigma)\} < \epsilon^*$$
(36)

for all  $\lambda \in (\lambda_4, 1)$ 

**Proof.** Under the event  $F \equiv \{\alpha_t, \beta_t, s_t\} \rightarrow (\delta, \delta, (D, D)\}$ , there must exist a date T such that  $\beta_T < \delta + 2b$ , where b is as given in Lemma 3. Since  $\beta_0 \geq \sigma$ , there exists  $\tilde{\lambda} < 1$  such that for any  $\lambda \in (\tilde{\lambda}, 1)$ , there must exist a first date k such that  $\beta_k \in (\delta + 2b, \sigma - 2b)$ . We assert that by permuting the two players if necessary, it must also be true that there exists a first date k in which simultaneously  $\beta_k \in (\delta + 2b, \sigma - 2b)$  and  $\alpha_k < \sigma - a$ , where a is also given by Lemma 3. Let  $E_k$  denote this event.

Take  $\epsilon = \frac{\epsilon^*}{3}$  and  $\lambda_4 = \max{\{\tilde{\lambda}, \lambda_1, \lambda_2, \lambda_3\}}$ , where  $\lambda_2, \lambda_3$  are as given in Lemmas 3 and 4 respectively, corresponding to this  $\epsilon$  and a, b, and  $\lambda_1$  is as given in Lemma 1 corresponding to this  $\epsilon$  and I(a, 2b). Taking any  $\lambda \in (\lambda_4, 1)$  and any k as defined above:

$$\begin{aligned} &\operatorname{Prob}^{\lambda}(F|E_{k},(\alpha_{0},\beta_{0})\geq(\sigma,\sigma)) \\ &= \operatorname{Prob}^{\lambda}(F|\alpha_{k}<\delta-a,\beta_{k}\in(\delta+2b,\sigma-2b))\operatorname{Prob}^{\lambda}(\alpha_{k}<\delta-a|E_{k},(\alpha_{0},\beta_{0})\geq(\sigma,\sigma)) \\ &+ \operatorname{Prob}^{\lambda}(F|(\alpha_{k},\beta_{k})\in N)\operatorname{Prob}^{\lambda}(\delta-a\leq\alpha_{k}\leq\delta+a|E_{k},(\alpha_{0},\beta_{0})\geq(\sigma,\sigma)) \\ &+ \operatorname{Prob}^{\lambda}(F|\delta+a<\alpha_{k}<\sigma-a,\delta+2b<\beta_{k}<\sigma-2b)\operatorname{Prob}^{\lambda}(F|E_{k},(\alpha_{0},\beta_{0})>(\sigma,\sigma)). \end{aligned}$$

where N is defined in Lemma 3. The first term on the RHS of this expression is bounded above by

$$\operatorname{Prob}^{\lambda}(\alpha_k < \delta - a | E_k, (\alpha_0, \beta_0) \ge (\sigma, \sigma))$$

Since the event we are conditioning on is a subset of the event  $\{\alpha_0 > \delta \ \beta_0 \le \theta\}$ . Lemma 4 implies that this term is less than  $\frac{\epsilon^*}{3}$ . The second term is bounded above by

$$\operatorname{Prob}^{\lambda}(F|(\alpha_k,\beta_k)\in N),$$

which is less than  $\frac{\epsilon^*}{3}$  by virtue of Lemma 3. Finally, the third term is bounded above by

$$\operatorname{Prob}^{\lambda}(F|(\alpha_k,\beta_k)\in I(a,2b))$$

which is also less than  $\frac{\epsilon^*}{3}$  by Lemma 1. Hence the sum of these terms is less than  $\epsilon^*$ , thereby completing the proof.

We are finally in a position to complete the

**Proof of Theorem 1.** Using Propositions 1 and 4, it suffices to show that in the 'untrembled' process, the probability of transiting from the (C, C) pss following a single tremble to any of the other three pss's converges to zero, while the probability of a reverse transition is bounded away from zero, as  $\lambda \to 1$ .

Suppose we start from the (C, D) pss, and player 1's aspiration experiences an upward tremble by x > 0, to take the aspiration vector to  $(x, \theta)$ . Now apply Lemma 3 (with  $b = \frac{x}{2}$  and  $a = \frac{x}{3}$ ) to infer that the probability of converging back to the (C, D) pss, converges to zero as  $\lambda \to 1$ . Moreover, the same Lemma also implies that (i) starting with the (C, D) pss followed with one tremble, the probability of converging to the (D, C) pss also converges to zero (since  $\beta_0 = \theta > \delta$ ), and (ii) the same is true for the probability of converging to the (D, D) pss or (C, C) pss and subject aspirations of one player to a single tremble. This establishes that as  $\lambda \to 1$ , the weight placed by  $\Pi^*$  on either the (C, D) pss or (D, C) pss must converge to zero.

Hence as  $\lambda \to 1$ , the sum of the weights placed on the (C, C) pss and the (D, D) pss will converge to 1.

Now suppose we start from the (D, D) pss, and player 2's aspiration experiences an upward tremble from  $\delta$  to  $\delta + x$ , while player 1's aspiration remains at  $\delta$ . The argument of Lemma 3 can then be applied (selecting a value of b smaller than x) to infer that from this state, the probability of the untrembled process converging to the (C, C) pss is close to 1, for  $\lambda$  sufficiently close to 1.

Finally, Lemma 5 shows that the probability of transiting to the (D, D) pss following application of one tremble to the (C, C) pss, converges to 0 as  $\lambda \to 1$ . This completes the proof.

### References

- Arthur, B. (1993), "On Designing Economic Agents that Behave like Human Agents," Journal of Evolutionary Economics 3, 1-22.
- Bendor, J., D. Mookherjee and D. Ray (1992), "Aspirations, Adaptive Learning and Cooperation in Repeated Games," Discussion Paper, Planning Unit Indian Statistical Institute, New Delhi. Revised, June 1995
- Binmore, K. and L. Samuelson (1992): "Evolutionary stability in games played by finite automata," *Journal of Economic Theory* 57, 278-305.
- Binmore, K. and L. Samuelson (1993), "Muddling Through: Noisy Equilibrium Selection," mimeo, University College, London
- Börgers, T. and R. Sarin (1994), "Learning Through Reinforcement and Replicator Dynamics," mimeo, University College London and Texas A & M University.
- Bush, R. and Mosteller, F. (1955), Stochastic Models of Learning. New York: John Wiley and Sons.
- Fudenberg, D. and E. Maskin (1990): "Evolution and cooperation in noisy repeated games," *American Economic Review* 80, 274-9.
- Gilboa, I. and D. Schmeidler (1992), "Case-Based Decision Theory," mimeo, Kellogg School of Management, Northwestern University.
- Kandori, M., G. Mailath and R. Rob (1993), "Learning, Mutation and Long Run Equilibria in Games," *Econometrica*, **61**(1), 29-56
- Kim, Y. (1995), "Satisficing, Cooperation and Coordination," mimeo, Department of Economics, Queen Mary and Westfield College, University of London.
- Meyn, S.P. and R.L. Tweedie (1993) Markov Chains and Stochastic Stability, London, New York: Springer-Verlag.
- Mookherjee D. and B. Sopher (1994), "Learning Behavior in an Experimental Matching Pennies Game," *Games and Economic Behavior*, 7, 62-91.

Sum Games," mimeo, Boston University.

- Pazgal A. (1995), "Satisficing Leads to Cooperation in Mutual Interests Games," mimeo, Kellogg School of Management, Northwestern University
- Robson, A. (1990): "Efficiency in evolutionary games: Darwin, Nash, and the secret handshake," Journal of Theoretical Biology 144, 379-96.

handshake," Journal of Theoretical Biology 144, 379-96

- Roth, A. and I. Erev (1993), "Learning in Extensive-Form Games: Experimental Data and Simple Dynamic Models in the Intermediate Term," forthcoming Games and Economic Behavior.
- Selten, R (1991), "Evolution, Learning and Economic Behavior." Games and Economic Behavior, 3(1).
- Suppes, P. and Atkinson, R. (1960). Markov Learning Models for Multiperson Interactions, Stanford University Press, Stanford.

Young P. (1993), "The Evolution of Conventions," Econometrica, 61(1), 57-84

### **PUBLISHED ISSUES\***

- WP-AD 92-01 "Inspections in Models of Adverse Selection" I. Ortuño. May 1992.
- WP-AD 92-02 "A Note on the Equal-Loss Principle for Bargaining Problems" C. Herrero, M.C. Marco. May 1992.
- WP-AD 92-03 "Numerical Representation of Partial Orderings" C. Herrero, B. Subiza. July 1992.
- WP-AD 92-04 "Differentiability of the Value Function in Stochastic Models" A.M. Gallego. July 1992.
- WP-AD 92-05 "Individually Rational Equal Loss Principle for Bargaining Problems" C. Herrero, M.C. Marco. November 1992.
- WP-AD 92-06 "On the Non-Cooperative Foundations of Cooperative Bargaining" L.C. Corchón, K. Ritzberger. November 1992.
- WP-AD 92-07 "Maximal Elements of Non Necessarily Acyclic Binary Relations" J.E. Peris, B. Subiza. December 1992.
- WP-AD 92-08 "Non-Bayesian Learning Under Imprecise Perceptions" F. Vega-Redondo. December 1992.
- WP-AD 92-09 "Distribution of Income and Aggregation of Demand" F. Marhuenda. December 1992.
- WP-AD 92-10 "Multilevel Evolution in Games" J. Canals, F. Vega-Redondo. December 1992.
- WP-AD 93-01 "Introspection and Equilibrium Selection in 2x2 Matrix Games" G. Olcina, A. Urbano. May 1993.
- WP-AD 93-02 "Credible Implementation"B. Chakravorti, L. Corchón, S. Wilkie. May 1993.
- WP-AD 93-03 "A Characterization of the Extended Claim-Egalitarian Solution" M.C. Marco. May 1993.
- WP-AD 93-04 "Industrial Dynamics, Path-Dependence and Technological Change" F. Vega-Redondo. July 1993.

<sup>\*</sup> Please contact IVIE's Publications Department to obtain a list of publications previous to 1992.

- WP-AD 93-05 "Shaping Long-Run Expectations in Problems of Coordination" F. Vega-Redondo. July 1993.
- WP-AD 93-06 "On the Generic Impossibility of Truthful Behavior: A Simple Approach" C. Beviá, L.C. Corchón. July 1993.
- WP-AD 93-07 "Cournot Oligopoly with 'Almost' Identical Convex Costs" N.S. Kukushkin, July 1993.
- WP-AD 93-08 "Comparative Statics for Market Games: The Strong Concavity Case" L.C. Corchón. July 1993.
- WP-AD 93-09 "Numerical Representation of Acyclic Preferences" B. Subiza. October 1993.
- WP-AD 93-10 "Dual Approaches to Utility" M. Browning. October 1993.
- WP-AD 93-11 "On the Evolution of Cooperation in General Games of Common Interest" F. Vega-Redondo. December 1993.
- WP-AD 93-12 "Divisionalization in Markets with Heterogeneous Goods" M. González-Maestre. December 1993.
- WP-AD 93-13 "Endogenous Reference Points and the Adjusted Proportional Solution for Bargaining Problems with Claims" C. Herrero. December 1993.
- WP-AD 94-01 "Equal Split Guarantee Solution in Economies with Indivisible Goods Consistency and Population Monotonicity"
   C. Beviá, March 1994.
- WP-AD 94-02 "Expectations, Drift and Volatility in Evolutionary Games" F. Vega-Redondo. March 1994.
- WP-AD 94-03 "Expectations, Institutions and Growth" F. Vega-Redondo. March 1994.
- WP-AD 94-04 "A Demand Function for Pseudotransitive Preferences" J.E. Peris, B. Subiza. March 1994.
- WP-AD 94-05 "Fair Allocation in a General Model with Indivisible Goods" C. Beviá. May 1994.
- WP-AD 94-06 "Honesty Versus Progressiveness in Income Tax Enforcement Problems"
   F. Marhuenda, I. Ortuño-Ortín. May 1994.

- WP-AD 94-07 "Existence and Efficiency of Equilibrium in Economies with Increasing Returns to Scale: An Exposition"
   A. Villar. May 1994.
- WP-AD 94-08 "Stability of Mixed Equilibria in Interactions Between Two Populations" A. Vasin. May 1994.
- WP-AD 94-09 "Imperfectly Competitive Markets, Trade Unions and Inflation: Do Imperfectly Competitive Markets Transmit More Inflation Than Perfectly Competitive Ones? A Theoretical Appraisal"
   L. Corchón, June 1994.
- WP-AD 94-10 "On the Competitive Effects of Divisionalization" L. Corchón, M. González-Maestre, June 1994.
- WP-AD 94-11 "Efficient Solutions for Bargaining Problems with Claims" M.C. Marco-Gil. June 1994.
- WP-AD 94-12 "Existence and Optimality of Social Equilibrium with Many Convex and Nonconvex Firms" A. Villar. July 1994.
- WP-AD 94-13 "Revealed Preference Axioms for Rational Choice on Nonfinite Sets" J.E. Peris, M.C. Sánchez, B. Subiza. July 1994.
- WP-AD 94-14 "Market Learning and Price-Dispersion" M.D. Alepuz, A. Urbano. July 1994.
- WP-AD 94-15 "Bargaining with Reference Points Bargaining with Claims: Egalitarian Solutions Reexamined" C. Herrero. September 1994.
- WP-AD 94-16 "The Importance of Fixed Costs in the Design of Trade Policies: An Exercise in the Theory of Second Best", L. Corchón, M. González-Maestre. September 1994.

- WP-AD 94-17 "Computers, Productivity and Market Structure" L. Corchón, S. Wilkie. October 1994.
- WP-AD 94-18 "Fiscal Policy Restrictions in a Monetary System: The Case of Spain" M.L. Escobedo, I. Mauleón. December 1994.
- WP-AD 94-19 "Pareto Optimal Improvements for Sunspots: The Golden Rule as a Target for Stabilization" S.K. Chattopadhyay. December 1994.
- WP-AD 95-01 "Cost Monotonic Mechanisms" M. Ginés, F. Marhuenda. March 1995.

- WP-AD 95-02 "Implementation of the Walrasian Correspondence by Market Games" L. Corchón, S. Wilkie. March 1995
- WP-AD 95-03 "Terms-of-Trade and the Current Account: A Two-Country/Two-Sector Growth Model" M.D. Guilló. March 1995.

States and the state of the states of the st

and the second of the second se

- WP-AD 95-04 "Exchange-Proofness or Divorce-Proofness? Stability in One-Sided Matching Markets" J. Alcalde. March 1995.
- WP-AD 95-05 "Implementation of Stable Solutions to Marriage Problems" J. Alcalde. March 1995.
- WP-AD 95-06 "Capabilities and Utilities" C. Herrero. March 1995.
- WP-AD 95-07 "Rational Choice on Nonfinite Sets by Means of Expansion-Contraction Axioms" M.C. Sánchez. March 1995.
- WP-AD 95-08 "Veto in Fixed Agenda Social Choice Correspondences" M.C. Sánchez, J.E. Peris. March 1995.
- WP-AD 95-09 "Temporary Equilibrium Dynamics with Bayesian Learning" S. Chatterji. March 1995.
- WP-AD 95-10 "Existence of Maximal Elements in a Binary Relation Relaxing the Convexity Condition" J.V. Llinares. May 1995.
- WP-AD 95-11 "Three Kinds of Utility Functions from the Measure Concept" J.E. Peris, B. Subiza. May 1995.
- WP-AD 95-12 "Classical Equilibrium with Increasing Returns" A. Villar. May 1995.
- WP-AD 95-13 "Bargaining with Claims in Economic Environments" C. Herrero. May 1995.
- WP-AD 95-14 "The Theory of Implementation when the Planner is a Player" S. Baliga, L. Corchón, T. Sjöström. May 1995.
- WP-AD 95-15 "Popular Support for Progressive Taxation" F. Marhuenda, I. Ortuño. May 1995.
- WP-AD 95-16 "Expanded Version of Regret Theory: Experimental Test" R. Sirvent, J. Tomás. July 1995.

- WP-AD 95-17 "Unified Treatment of the Problem of Existence of Maximal Elements in Binary Relations. A Characterization"
   J.V. Llinares. July 1995.
- WP-AD 95-18 "A Note on Stability of Best Reply and Gradient Systems with Applications to Imperfectly Competitive Models" L.C. Corchón, A. Mas-Colell. July 1995.
- WP-AD 95-19 "Redistribution and Individual Characteristics" I. Iturbe-Ormaetxe. September 1995.
- WP-AD 95-20 "A Mechanism for Meta-Bargaining Problems" M<sup>a</sup>. C. Marco, J. E. Peris, B. Subiza. September 1995.
- WP-AD 95-21 "Signalling Games and Incentive Dominance" G. Olcina, A. Urbano. September 1995.
- WP-AD 95-22 "Multiple Adverse Selection" J.M. López-Cuñat. December 1995.
- WP-AD 95-23 "Ranking Social Decisions without Individual Preferences on the Basis of Opportunities" C. Herrero, I. Iturbe, J. Nieto. December 1995.
- WP-AD 95-24 "The Extended Claim-Egalitarian Solution across Cardinalities" M<sup>a</sup> C. Marco. December 1995.
- WP-AD 95-25 "A Decent Proposal" L. Corchón, C. Herrero. December 1995.
- WP-AD 96-01 "A Spatial Model of Political Competition and Proportional Representation" I. Ortuño. February 1996.
- WP-AD 96-02 "Temporary Equilibrium with Learning: The Stability of Random Walk Beliefs" S. Chatterji. February 1996.
- WP-AD 96-03 "Marketing Cooperation for Differentiated Products" M. Peitz. February 1996.
- WP-AD 96-04 "Individual Rights and Collective Responsibility: The Rights-Egalitarian Solution" C.Herrero, M. Maschler, A. Villar. April 1996.

- WP-AD 96-05 "The Evolution of Walrasian Behavior" F. Vega-Redondo April 1996
- WP-AD 96-06 "Evolving Aspirations and Cooperation"R. Karandikar, D. Mokherjee, D. Ray, F. Vega-Redondo. April 1996.